Author's response to reviews

Title: Improving education in primary care: development of an online curriculum using the blended learning model

Authors:

Linda Orkin Lewin (llewin@peds.umaryland.edu)
Mamta Singh (mamta.singh@case.edu)
Betzi L. Bateman (b.bateman@csuohio.edu)
Pamela Bligh Glover (pamela.glover@case.edu)

Version: 5 Date: 4 June 2009

Author's response to reviews:

The following changes were made in this proofing stage:

Typographical and grammatical changes were made based on spell check as below:

1. Betzi Bateman’s current address: It was updated to: Department of Educational Foundations and Special Services, School of Lifespan Development and Educational Services, Kent State University, Kent, Ohio

Pager 4: last sentence: add the word "The" at the beginning so it reads "The blended learning model is a useful tool..."

Page 8: Top paragraph: third sentence: add the word "original" so it reads, "... so the original CPCP modules used open-ended questions..."

Page 8: second paragraph: third line from bottom add "were required to" so sentence reads, "Students were required to submit a SOAP note from..."

Page 12: a period is missing in the second line after "majority of students."

Page 18: acknowledgements: take out words "assistance with her" so it reads "We thank Dr. Klara Papp for her statistical support."

Page 20-22: in references we changed the following ref numbers to sentence case from title case: we removed the capitals in the titles of the papers in numbers 5,6,9,13,14,18,19, and 21.

No changes were made to content.